Reviewer #1 and #2:
Thank you for your review.

Reviewer #3 (Alexander N Kharlamov)

Dear author, The article represents the review aiming to provide a general overview of PPM/ICD function, review documented EMI effect on these devices, and acknowledge current management of CIED interference. The article is written with good English-speaking adduction of the arguments. The article is sufficiently novel and very interesting to warrant publication. All the key elements are presented and described clearly. The most discussable options in the article are: 1) Please correct all your minor grammar errors and typos. 2) I would generally suggest to draw a table with Guidelines-style findings to navigate physicians with some daily practice recommendations.

1. I have corrected all grammatical errors.
2. I have included an additional table (Table 3) in the PERI-PROCEDURAL MANAGEMENT OF CIED INTERFERENCE: section summarizing suggested CIED peri-procedural settings.